Author’s response to reviews

Title: Impact of sarcomatoid differentiation and rhabdoid differentiation on prognosis for renal cell carcinoma with vena caval tumour thrombus treated surgically

Authors:

Bin Yang (yangbin422@163.com)

Haizhui Xia (xiahaizhui@bjmu.edu.cn)

Chuxiao Xu (xuchuxiao@bjmu.edu.cn)

Min Lu (lumin@bjmu.edu.cn)

Shudong Zhang (zhangshudong@medmail.com.cn)

Guoliang Wang (wangguoliang@medmail.com.cn)

Lulin Ma (malulin@bjmu.edu.cn)

Version: 5 Date: 29 Jan 2020

Author’s response to reviews:

1. Authors contribution

Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Authors' Contributions

LLM, GLW proposed the protocol. BY, HZX, CXX, ML were involved in data collection and management. BY, HZX, SDZ analysed the data. BY, HZX contributed to statistical analysis. BY contributed to manuscript writing. LLM, GLW critically revised the manuscript.

2. Tables

Table 1 seems to be missing however it is cited within the text, please clarify this.

Table 1 has been supplemented in the revision.

3. Clean copy

The latest revision is a clean copy.